Author's response to reviews

Title: Pregnancy And Liver adenoma Management: PALM-study

Authors:

Sanne M van Aalten (s.vanaalten@erasusmc.nl)
Mirelle E.E. Bröker (m.broker@erasusmc.nl)
Jan J.V. van Busschbach (j.vanbusschbach@erasusmc.nl)
Harry J. de Koning (h.dekoning@erasusmc.nl)
Robert A. de Man (r.deman@erasusmc.nl)
Eric A.P. Steegers (e.a.p.steegers@erasusmc.nl)
Ewout W. Steyerberg (e.steyerberg@erasusmc.nl)
Turkan Terkivatan (t.terkivatan@erasusmc.nl)
Jan N.M. IJzermans (j.ijzermans@erasusmc.nl)

Version: 2 Date: 4 May 2012

Author's response to reviews: see over
To the Editor

BMC gastroenterology

Subject: Pregnancy And Liver adenoma Management: PALM-study

MS: 7084636547150724

Rotterdam, May 4th 2012

Hereby we would like to resubmit the above named manuscript for consideration in your journal.

Attached please find our proofs of ethics and funding proofs. We received funding from the fonds NutsOhra and stichting Coolsingel in the Netherlands. Our ethical proofs from our ethical board are attached and registered with number NL36058.078.11.

Sincerely,
On behalf of all authors,

Jan IJzermans

Corresponding author
Prof. dr. J.N.M. IJzermans
Department of Surgery
Erasmus Medical Center
PO Box 2040
3000 CA Rotterdam, the Netherlands
Tel: +31-10- 704 3727